Author's response to reviews

**Title:** Upper limbs dysmetria caused by cervical spinal cord injury: a case report

**Authors:**

Hsun-Chang Lin (dokugawa0502@hotmail.com)
Chun-Hung Chen (abcd1234@hotmail.com)
Gim-Thean Khor (respiratory5566@hotmail.com)
Poyin Huang (Hamamatsu0502@hotmail.com)

**Version:** 3  **Date:** 7 September 2009

**Author's response to reviews:** see over
Dear Editor:

We sincerely appreciated the Reviewer’s comments. Point-by-point responses were provided and the manuscript was revised as well.

1. The manuscript was revised according to the journal style. The consent was obtained from the patient. The statement “Copy paste” was also added to the Acknowledgement (?) section.
2. Figure legend and figure were separated according to the journal style.

Poyin Huang